Author’s response to reviews

Title: Antipsychotic polypharmacy in a regional health service: a population-based study

Authors:

Miguel Bernardo (bernardo@clinic.ub.es)
Anna Coma (acomaf@catsalut.cat)
Cristina Ibañez (cibanezc@catsalut.cat)
Corinne Zara (czara@catsalut.cat)
Josep Maria Bari (jbari@clinic.ub.es)
Antoni Serrano-Blanco (aserrano@pssjd.org)

Version: 7 Date: 2 February 2012

Author’s response to reviews: see over
To the Executive Editor
BioMed Central
Dear Ms Majithia,
Thank you for the revision you sent us on the manuscript we have re-submitted. As recommended, we have copy-edited the whole text, revising and improving the style of written English. We hope to fulfil the requirements for acceptation and publication. In case there are any other suggestions, please contact us at your earliest convenience.
Sincerely,

Miguel Bernardo, MD, PhD